Author’s response to reviews

Title: A mixed methods evaluation of an integrated adult mental health service model

Authors:
Diana Beere (dianabeere@gmail.com)
Imogen Page (i.page@qcmhr.uq.edu.au)
Sandra Diminic (sandra_diminic@qcmhr.uq.edu.au)
Meredith Harris (meredith_harris@qcmhr.uq.edu.au)

Version: 2 Date: 25 Jul 2019

Author’s response to reviews:

Dear Editor,

Thank you for the opportunity to submit a revised manuscript. We have made the necessary changes as detailed in your Decision Letter and have submitted a revised manuscript and figure files.

Kind regards,

Imogen Page

(on behalf of all authors)